Author's response to reviews

Title: Tobacco use among students aged 13-15 years in Greece: the GYTS project

Authors:

  Athina Kyrlesi (Kyrlesi@med.uth.gr)
  Elpidoforos S. Soteriades (esoteria@cyprusinstitute.org)
  Charles W. Warren (wcw1@cdc.gov)
  Jeni Kremastinou (jkrem@forthnet.gr)
  Panagiotis Papastergiou (papastergiou@med.uth.gr)
  Nathan R. Jones (naj5@cdc.gov)
  Christos Hadjichristodoulou (xhatzi@med.uth.gr)

Version: 4  Date: 5 January 2007

Author's response to reviews:

Dear Editor,

Please find attached the revised version of the manuscript titled "Tobacco use among students aged 13-15 years in Greece: the GYTS project". In this revision we included all the formatting changes requested by the Editor.

Yours
Christos Hadjichristodoulou